Author's response to reviews

Title: Prevalence of HIV among MSM in Europe: comparison of self-reported diagnoses from a large scale internet survey and existing national estimates

Authors:

Ulrich Marcus (MarcusU@rki.de)
Ford Hickson (ford.hickson@emis-project.eu)
Peter Weatherburn (peter.weatherburn@emis-project.eu)
Axel J. Schmidt (axel.j.schmidt@emis-project.eu)

Version: 2 Date: 27 June 2012

Author's response to reviews:

Dear editors,

According to your request from June 27, 2012 we have revised the manuscript by
1) Including a reference to the Ethics Committee approval in the Methods section on page 6 (end of first paragraph)
2) Including a Competing interest statement on page 14 (between Authors contributions and References).

Sincerely
Ulrich Marcus